

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Antman, Francisca M.; Cortes, Kalena E.

Working Paper The Long-Run Impacts of Mexican-American School Desegregation

IZA Discussion Papers, No. 15019

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Antman, Francisca M.; Cortes, Kalena E. (2022) : The Long-Run Impacts of Mexican-American School Desegregation, IZA Discussion Papers, No. 15019, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/250680

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15019

The Long-Run Impacts of Mexican-American School Desegregation

Francisca M. Antman Kalena E. Cortes

JANUARY 2022



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15019

The Long-Run Impacts of Mexican-American School Desegregation

Francisca M. Antman University of Colorado Boulder and IZA

Kalena E. Cortes Texas A&M University and IZA

JANUARY 2022

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

	IZA – Institute of Labor Economics	
Schaumburg-Lippe-Straße 5–9 53113 Bonn, Germany	Phone: +49-228-3894-0 Email: publications@iza.org	www.iza.org

ABSTRACT

The Long-Run Impacts of Mexican-American School Desegregation*

We present the first quantitative analysis of the impact of ending *de jure* segregation of Mexican-American school children in the United States by examining the effects of the 1947 *Mendez v. Westminster* court decision on long-run educational attainment for Hispanics and non-Hispanic whites in California. Our identification strategy relies on comparing individuals across California counties that vary in their likelihood of segregating and across birth cohorts that vary in their exposure to the *Mendez* court ruling based on school start age. Results point to a significant increase in educational attainment for Hispanics who were fully exposed to school desegregation.

JEL Classification:	I24, I26, J15, J18
Keywords:	Mexican-American, school desegregation, Mendez v.
	Westminster

Corresponding author:

Francisca M. Antman Department of Economics University of Colorado Boulder 256 UCB, Boulder CO 80309 USA E-mail: francisca.antman@colorado.edu

^{*} We thank Richard Akresh, Elizabeth Cascio, Ellora Derenoncourt, Monica Deza, Rubén Donato, Myron Gutmann, Nate Hilger, Gerald Jaynes, Trevon Logan, Terra McKinnish, Tomás Monarrez, Martha Olney, John Parman, Christina Romer, and seminar participants at the University of California Berkeley, meetings of the American Economic Association, Economic History Association, Population Association of America, Western Economic Association International, NBER's Working Group on Race and Stratification, as well as the Race, Ethnicity, Gender, and Economic Justice Virtual Symposium for helpful feedback. Stephanie Gullo, Jacob Kirsch, Fahad Manzoor, and Rimjhim Saxena provided excellent research assistance. Institutional support from Texas A&M University and development support from the University of Colorado Boulder Population Center (CUPC) are also gratefully acknowledged. Research reported in this publication was supported by the Eunice Kennedy Shriver National Institute of Child Health & Human Development of the National Institutes of Health under Award Number P2CHD066613. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

"In response to the demand, a small frame building was erected behind the main elementary school building. Here all the Mexican and Mexican-American children were housed."

--Elis Tipton, Principal of the San Dimas Elementary School during the 1940s. (As cited in Hendrick, 1975, p. 207)

I. INTRODUCTION

The rise in majority-minority communities driven by demographic trends and residential segregation have again placed public attention on the issue of school segregation, with a new focus on Latinx groups who are now among the most segregated in states like California (Orfield and Ee 2014).¹ Although the driving force of current school segregation outcomes may be regarded as *de facto* in nature, much can be learned from the impacts of eliminating *de jure* segregation, as with the extensive studies of the impact of court-ordered desegregation in the U.S. South on the education of African-Americans (Guryan 2004; Johnson 2011).² At the same time, the literature has paid relatively little attention to Mexican segregation, despite historical accounts which point to widespread segregation of Mexican-Americans throughout the southwestern United States in the early to mid-20th century, and the local establishment of so-called "Mexican schools" (Wollenberg 1976).³ Our paper provides the first quantitative analysis of the impact of ending *de jure* segregation of Mexican-American school children by examining the effects of the 1947 *Mendez v. Westminster* decision on long-run educational attainment of Hispanics in California.

The *Mendez* decision, which originated as a class-action suit challenging Mexican school segregation by parents of Mexican children in Orange County, California, is often described as an important precursor to the landmark 1954 case, *Brown v. Board of Education* (Meraji 2014; Kandil 2016). Indeed, the *Mendez* decision was the first successful constitutional challenge to segregation in federal courts on the basis that segregation of Mexican students violated the equal protection clause of the 14th Amendment, an argument

¹ Logan and Parman (2017) trace the dramatic rise in residential segregation across the U.S.

² We use the terms African-American and Black interchangeably throughout as well as the terms Latinx, Latino, and Hispanic interchangeably throughout

³ We use the terms Mexican-American and Mexican interchangeably to refer to individuals of Mexican ethnic origin, regardless of citizenship.

that would reemerge in the *Brown* case several years later (Valencia 2005). However, just as prior studies have suggested that the full legal impacts of *Brown* were not felt until 10 years later (Reber 2005), the question of whether the end of *de jure* Mexican segregation brought about by *Mendez* ultimately led to improvements in educational attainment for Mexican-Americans is an empirical one which we investigate here.

One important difference between examining the impact of desegregation of Mexican children in California also suggests why the subject has been underexamined in the literature. Unlike some states in the American South, where official documents differentiate between schools for Black and white children, segregation policy in California was primarily decided at the local level, and though widespread in certain areas, official documentation of the extent and geographic patterns of segregation are limited in comparison. Nevertheless, historical sources suggest that segregated schools were prevalent in areas with relatively high shares of Hispanic populations (Wollenberg 1976; Bowman 2001). Thus, we leverage an intent-to-treat difference-in-differences research design that compares counties with a high Hispanic to non-Hispanic population ratio in 1940 to counties with a relatively low ratio, and across birth cohorts that started school after Mendez relative to birth cohorts that started school before the *Mendez* ruling. Analogous to those studies which have examined the impacts of school desegregation for African-Americans (see for example, Guryan 2004; Johnson 2011; Reber 2010; Card and Kruger 1992; Ashenfelter, Collins, and Yoon 2006), we examine educational attainment (completed schooling), but our primary focus is the Hispanic population. To provide further weight to our identification strategy, we check for parallel trends prior to treatment impacts by way of a placebo experiment on the birth cohorts who should have been unaffected by Mendez based on their school start age. We also trace out impacts by birth year across counties well before and after the *Mendez* ruling to see the full extent of desegregation impacts on cohorts across treated and comparison areas based on school start age.

Our results suggest that the impact of school desegregation in 1947 was significant, leading to an increase of almost 0.9 years of schooling for Hispanics in the cohorts who started school after *Mendez* relative to cohorts born 10 years prior. As the latter group may have been partially affected by *Mendez*, we also estimate impacts relative to an older cohort, born 10-20 years prior, and find even larger effects on the order of 1.9 years of schooling,

3

which can be compared with an average of 9.2 years for the oldest birth cohorts. Finally, we examine impacts on non-Hispanic whites using the same identification strategy and find evidence of a decline in educational attainment of non-Hispanic whites in birth cohorts that began school after *Mendez* in counties that were more likely to be segregated, relative to those cohorts who began school prior to *Mendez* in counties that were less likely to be segregated. Given that these changes occurred relatively quickly, these facts point to the likelihood of a movement toward a more equitable resource distribution within counties across students from different ethnic groups following the end of *de jure* segregation.⁴ However, in the absence of data on school resources explicitly designated as "Mexican" or "white" within counties, we cannot decisively pin down the mechanism. Nevertheless, our results suggest an important causal link between desegregation and greater equity across students of different backgrounds.

The remainder of this paper proceeds as follows. Section II provides background on the history of Mexican-American segregation in the Southwest and the *Mendez* court decision. Section III summarizes the data set used and Section IV describes in detail the empirical strategy. Section V presents the results and Section VI concludes with a discussion of potential mechanisms and avenues for future research.

II. MEXICAN SCHOOL SEGREGATION AND THE MENDEZ V. WESTMINSTER DECISION

The roots of segregation in the American Southwest can be traced back as far as the mid-nineteenth century when the territory was annexed by the United States at the close of the Mexican-American war (Valencia 2005).⁵ However, the rise of segregated schools in the region is typically linked to the increased waves of Mexican migration to the American Southwest in the early 20th century and the coinciding rise in prejudice toward individuals of Mexican origin (Wollenberg 1976). By 1927, nearly 10 percent of the state's school children were of Mexican descent, and in some counties, they were closer to 20 or 30 percent

⁴ School resources have been an important avenue of exploration in the literature on segregation in the South and its impact on African-American educational outcomes (Carruthers and Wannamaker 2017; Cascio et al. 2010; Margo 1990).

⁵ While California was once governed by Mexico, Mexicans who remained there at the conclusion of the Mexican-American war were granted full U.S. citizenship under the Treaty of Guadalupe Hidalgo (1848, Articles VIII and IX).

of the total school population (Wollenberg 1976, p. 110-11). As for African-Americans in the U.S. South, an underlying pattern of residential segregation emerged that was often codified into law and reinforced through violence and harassment (Donato, Menchaca, and Valencia 1991; Cook, Logan, and Parman 2017). School boundaries were often drawn so that Mexican children could only attend the local Mexican school, and white children were granted transfers out of the Mexican school zone, thus solidifying segregation in the schools (Gonzalez 1985).

However, it is important to note that segregation in Mexican schools was never universally implemented in the same way throughout California. The practice was not part of the state's education code and thus its implementation varied at the local level with some school districts establishing separate schools (Hendrick 1975) or classrooms, or choosing not to separate students at all, or segregating only younger children (Wollenberg 1976). This variation in treatment status, and the failure to document the extent of segregation on a large scale,⁶ make economic analysis of the impact of Mexican desegregation challenging and thus necessitate the use of the intent-to-treat research design we pursue here.⁷ Nevertheless, by 1931, 85 percent of California schools reported segregating Mexican students at the classroom or school level (Hendrick 1977, as cited in Donato, Menchaca, and Valencia 1991), thus underscoring the fact that segregation was widespread throughout California.

The justification for segregation was often rooted in racial discrimination which was sometimes spelled out explicitly, as with those who wished to characterize Mexicans as "Indians," a group which could be legally segregated at the time (Hendrick 1977, as cited in Donato, Menchaca, and Valencia 1991). Similar arguments were made on the basis of cleanliness, intellectual ability, and the supposed inferiority of Mexicans who were deemed to require special instruction to acculturate to American standards (Carter 1970). Economic interests also played a role in segregation as employers resisted offering education that would jeopardize their primary source of cheap labor (Gonzalez 1985). One of the more salient legal justifications was that language deficiencies on the part of Mexican children

⁶ Indeed, Hendrick (1975, p. 213) describes how Los Angeles school officials explicitly stopped recording the race of students and employees.

⁷ This marks an important distinction between segregation of African-Americans in the South and the Mexican experience in the Southwest.

required segregation to improve their English language skills, despite the fact that Mexican-American children who did not speak Spanish were also segregated (Donato, Menchaca, and Valencia 1991). Nevertheless, the underlying motivation was clear to the extent that segregation in schools mirrored the society at large where barriers commonly existed to separate Mexicans and whites in facilities such as parks, movie theaters, and swimming pools (Arriola 1995; Kandil 2016).

In this context, the struggle to desegregate schools began in the 1930s in Texas and California when the legality of segregating Mexican school children began to be challenged in the courts to varying success at the local level (for a review see Valencia 2005). In 1945, Gonzalo Mendez and four other Mexican-American parents sued four Orange County school districts in federal court on behalf of their children and 5,000 other children of "Mexican and Latin descent," arguing that segregation violated their constitutional rights (Valencia 2005; *Mendez* 1947). Unlike prior suits, *Mendez* gained significant attention for being the first to claim that the doctrine of "separate but equal" violated the equal protection clause of the 14th Amendment (Gonzalez 1990)—an argument that would resurface in the 1954 *Brown* decision.

However, there are reasons to believe the impact of *Mendez* stopped short of the legal precedent later to be set by *Brown*. First, an injunction against Mexican segregation was issued in a Federal District court and while the defendants initially resisted its implementation, the ruling was ultimately upheld at the Ninth U.S. Circuit Court of Appeals in 1947 (Gonzalez 1985). The case never reached the Supreme Court because the school district declined to appeal, thus leading some to believe that the ruling might only be locally relevant (Arriola 1995). Second, segregation was ultimately overturned in *Mendez* based on its opposition to existing California law (*Westminster* 1947), rather than a direct rejection of the doctrine of "separate but equal" espoused in *Plessy v. Ferguson* (Valencia 2005). Nevertheless, the controversy highlighted in *Mendez* led legislators to swiftly repeal any remaining school segregation laws in 1947 (Wollenberg 1976), with the support of Governor Earl Warren, who would later serve as Supreme Court justice during the *Brown* decision.⁸

⁸ The participation of the National Association for the Advancement of Colored People (NAACP) and Thurgood Marshall, who helped argue against the legality of segregation in the *Mendez* case, also mark important linkages

Thus, it is reasonable to believe that, in the wake of the *Mendez* decision, California school officials would have understood segregation to be in violation of the law and historical surveys taken at the time suggest that integration efforts were immediately put into place (Gonzalez 1985; Wollenberg 1976).

III. DATA SOURCE AND DESCRIPTIVE STATISTICS

The main data set used in this analysis comes from the five percent Public Use Samples of the 1980, 1990, and 2000 U.S. decennial censuses as made available through the Integrated Public Use Microdata Series (IPUMS). Throughout portions of the analysis, we consider Hispanic and non-Hispanic white individuals who were born in and reside in California, with birth years spanning 1921-1945, where Hispanics are individuals who identify as Hispanic/Latino, regardless of race. Non-Hispanic whites are individuals who identify as white on the race question, but not as Hispanic/Latino.⁹

We emphasize that no one in our sample is an immigrant, and the birth cohorts in our sample pre-date the large waves of immigration from Mexico and Latin America seen toward the end of the 20th century. Nevertheless, some individuals in our sample may certainly be the children, grandchildren, or great-grandchildren of immigrants. Since higher-order immigrant generations have generally been found to have worse educational outcomes than non-Hispanic whites (Duncan and Trejo 2011), if any immigrant effects were coupled with the impacts of segregation that we measure here, we would expect to see lower educational progress of Hispanics in our sample. Similarly, to the extent that any Hispanics by heritage are identifying as non-Hispanic whites in our sample, we would also expect to see lower educational progress for Hispanics vis-à-vis non-Hispanic whites, since so-called "ethnic attritors" have been found to have characteristics closer to the non-Hispanic white population (Duncan and Trejo 2011; Antman, Duncan, and Trejo 2016, 2020). Thus, both immigrant assimilation and ethnic attrition would work against us finding any positive impacts of desegregation.

with *Brown*. Marshall would later serve as counsel for the African-American plaintiffs in *Brown* before the Supreme Court (Valencia 2005).

⁹ Individuals identifying with other racial groups are dropped from the sample. Unfortunately, the group of non-Hispanic Blacks in the sample is too small for meaningful analysis.

Note that our data restrictions imply that, at the time they are surveyed, all individuals in our sample are well beyond the age when they would likely have completed schooling. Also note that we include a broader classification of Hispanics, rather than restricting attention exclusively to Mexican-Americans, because it is believed that, in general, all individuals identifying as Hispanic would have been treated similarly to Mexican-Americans in the era of segregation.¹⁰ Moreover, focusing on Hispanics born in California in the 1920s through 1940s, as we do here, necessarily excludes immigrants from Central and South America that arrived in the U.S. in the latter half of the 20th century. Thus, it is reasonable to expect that our Hispanic sample consists primarily of individuals of Mexican descent.¹¹

To identify likely segregated areas, we use data from the complete count 1940 U.S. Census as made available through the IPUMS (Ruggles, et al. 2015).¹² Since there are no official records of which areas were separating Mexican and Mexican-American students into separate schools and classrooms, we leverage historical accounts which indicate segregation was most prevalent in counties with a high share of Hispanics in the population (Wollenberg 1976, Bowman 2001).¹³ To make these notions concrete, we calculate the Hispanic to non-Hispanic population ratio by county for all California counties in 1940 and define counties as having a high likelihood of being segregated (high segregation, henceforth *HiSeg*) if they are above the 75th percentile level of this distribution (approximately 0.09). For comparison,

¹⁰ Indeed, the plaintiffs in *Mendez* asserted that segregated schools were attended by children of "Mexican and Latin descent" (*Mendez* 1947). Focusing the empirical investigation on the larger group of Hispanics also avoids the issue of whether Mexican-Americans would identify themselves as Mexican/Mexican-American or Hispanic, *per se*.

¹¹ Even today, Mexican/Mexican-Americans are overwhelmingly the major demographic origin group within the set of individuals identifying as Hispanic (Motel and Patten 2012).

¹² For robustness, we have also used the 1930 Census to define our high- and low-likelihood segregation counties. The advantage of this approach is that "Mexican" was listed as a race category in 1930, however, we prefer to use the 1940 Census because it is the closest to the period prior to *Mendez*. Event study analyses using the 1930 Census categories are shown in appendix Figures A3 through A8.

¹³ An alternative approach might use a measure based on the absolute number of Hispanics as an indicator of the extent of likely segregation. We have tried that approach and find similar results, but the sample in the comparison group (low segregation counties) drops some, likely due to the fact that those are low population counties and thus less likely to appear or be identified in the public use sample. The other drawback of using a measure of likely segregation that is based on high absolute numbers of Hispanics is that it is likely to be highly correlated with an indicator for high overall population, and as a result some would view the subsequent analysis as amounting to a comparison of educational attainment in very high versus very low population areas over time. Consequently, this alternative approach would be more problematic, and thus we prefer the segregation measure described below, which also accords with historical accounts suggesting that the share of Hispanic individuals was relevant for determining segregation.



Figure 1: Educational Attainment in High and Low Segregated Counties by Age at *Mendez v. Westminster* School Desegregation Decision

Notes: Sample is limited to men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1940 counties: high segregation, HiSeg), or very low (i.e., below the 25% level for all 1940 counties: low segregation, LoSeg).

we define counties as having a low likelihood of being segregated (low segregation, henceforth *LoSeg*) if they are below the 25th percentile level in the same distribution (about 0.02)¹⁴, and drop counties which lie between the 25th and 75th percentiles from the analysis.¹⁵ Note that our definitions are consistent with historical evidence on segregation: Orange County, from which the *Mendez v. Westminster* case originated, is in the *HiSeg* county group, as well as the counties associated with those towns having segregated schools which are cited in historical sources (Bowman 2001).¹⁶ Our own calculations using California State Department of Education (1932) records also indicate that the percent of registered minors who were classified as Mexican was significantly higher in counties classified as *HiSeg* versus *LoSeg* in our main sample (17.58 percent for *HiSeg* and 0.95 percent for *LoSeg*, on average).¹⁷

As a first step of descriptive analysis, Figure 1 traces out the average educational attainment by birth year for Hispanics and non-Hispanic whites residing in *HiSeg* and *LoSeg* counties.¹⁸ Several interesting points are illuminated from the figure, to be investigated in the regression analysis below. First, the dramatic rise in the educational attainment of Hispanics in *HiSeg* counties can be observed for students who were school age or younger at the time of the *Mendez* decision. At the same time, for older Hispanics in *HiSeg* counties, who were already in their twenties at the time of *Mendez*, educational attainment hovers around 9 years of education on average. In comparison, Hispanics in *LoSeg* counties have educational attainment closer to 12 years on average, and hovers around that level for all cohorts, before

¹⁴ For robustness, we have also used the upper and lower thirds of the distribution as the thresholds defining high- and low-likelihood segregation counties and have found similar results (see appendix Tables A3 and A4). Note that this robustness check implies a different set of counties than the main results, and Los Angeles is included in the High Segregation counties in this alternative set.

¹⁵ For the main analytical sample, high segregation (*HiSeg*) counties include: Fresno, Kings, Madera, Merced, Monterey, Orange, Riverside, San Bernadino, San Luis Obispo, Santa Barbara, Tulare, Ventura, and Yolo. Low segregation (*LoSeg*) counties include: Butte, El Dorado, Humboldt, Shasta, and Sonoma. More *HiSeg* counties are identified than *LoSeg* counties because the empirical strategy uses the complete count 1940 Census to make this distinction, but *LoSeg* counties are less likely to be named in the public use 1980-2000 Census samples.

¹⁶ Two notable exceptions are Los Angeles and San Diego counties where the Hispanic-to-non-Hispanic population ratios fall just below the *HiSeg* level, and are thus dropped from the main analysis here, for consistency with the empirical method. This suggests that our main point estimates are likely to be conservative, relative to what we would obtain if we were able to measure segregation directly. However, as noted elsewhere, Los Angeles and San Diego counties are included in the sample of high segregation counties in two of our robustness checks, in which we find similar results.

¹⁷ While California school records document the existence of other types of schools which Mexican-American children may have attended (e.g., migratory schools), such records predate the analysis and cohorts used in this paper.

¹⁸ Note that all individuals in our sample were also born in California, but to our knowledge, more precise information on place of birth is not available from this or any large-scale data set for these birth cohorts.

	Panel / 1941- (Ages 2-	A: Treatme 1945 Birth 6 at Mende	ent Group, 1 Cohorts 22 Decision)	Panel F 1931 (Ages 12-	3: Comparis 1935 Birth 16 at Menc	son Group, Cohorts lez Decision)	
	Segreg	gation:		Segre	gation:		
	High	Low	$\Delta^{treatment}$	High	Low	$\Delta^{comparison}$	$\Delta^{treatment} - \Delta^{comparison}$
Years of Education	11.705 [2.808]	12.797 [2.530]	-1.092*** (0.217)	10.228 [3.736]	12.026 [2.248]	-1.798*** (0.267)	0.706** (0.344)
Jr. High (8+ years of education)	0.932 [0.251]	0.979 [0.144]	-0.047*** (0.013)	0.792 [0.406]	0.974 [0.160]	-0.182*** (0.020)	0.135*** (0.024)
High School (12+ years of education)	0.673 [0.469]	0.811 [0.393]	-0.138*** (0.034)	0.482 [0.500]	0.701 [0.461]	-0.219*** (0.053)	0.081 (0.063)
Number of observations	3,237	143	3,380	2,164	77	2,241	5,621

Table 1: Sample Descriptive Statistics on Mexican-American Educational Attainment by Intent-to-Treat Status

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 75% level of all California counties based on 1940 full-count Census. Low segregation (LoSeg) indicates that the Hispanic to non-Hispanic population ratio is below the 25% level of all California counties based on 1940 full-count Census. Sample is limited to men and women from 5% samples of 1980, 1990, and 2000 Censuses with birth cohorts between 1941 and 1945 (treatment group) or between 1931 and 1935 (comparison group). Standard deviations in brackets. Robust standard errors in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10.

and after *Mendez*. This constitutes suggestive evidence that Hispanics in segregated counties were the main beneficiaries of the *Mendez* decision and, though these are only summary statistics, also provide some suggestive support for the parallel trends assumption behind the intent-to-treat difference-in-differences research design, to be assessed more rigorously below.

Figure 1 also shows that non-Hispanic whites have higher educational attainment than Hispanics in both the segregated and non-segregated counties for all cohorts. But the gap between non-Hispanic whites and Hispanics from older cohorts is substantially larger in *HiSeg* counties compared with the analogous gap in *LoSeg* counties. Moreover, while Hispanics in *LoSeg* counties have substantially higher years of education than Hispanics in *HiSeg* counties, this pattern is reversed for non-Hispanic whites, who have the highest educational attainment in *HiSeg* counties except for in the cohorts that were fully exposed to desegregation. These details provide suggestive support for the hypothesis that older Hispanics in *HiSeg* counties were profoundly disserved by the educational system available to them, which would be consistent with a segregated school system that advantaged non-Hispanic whites in *HiSeg* counties over earlier periods.

It is also noteworthy that the gap between non-Hispanic whites in *HiSeg* versus *LoSeg* counties is much smaller than that between Hispanics in *HiSeg* and *LoSeg* counties. This would also suggest the existence of a more consistent school system in place for non-Hispanic whites across counties, and thus segregated schools for Hispanics in *HiSeg* counties. Finally, the gap between non-Hispanic whites in *HiSeg* and *LoSeg* counties is relatively small throughout the cohorts observed, with a small increase in educational attainment in *LoSeg* counties relative to *HiSeg* evident for the youngest cohorts who would have been fully exposed to *Mendez*. This would be consistent with resources being shifted toward Hispanics and away from non-Hispanic whites in *HiSeg* counties after *Mendez* resulting in a decline in their relative advantage. This will be explored in subsequent analysis where we investigate the impacts of desegregation on non-Hispanic whites as an extension below.

To make these differences concrete, Table 1 describes the main Hispanic sample and illustrates the difference-in-differences empirical strategy in summary statistics for the outcomes of interest, using the primary treatment and comparison groups. Consistent with the evidence from Figure 1, we see a large difference in completed education and likelihood

12

of junior high and high school completion rates across *HiSeg* and *LoSeg* counties. For older cohorts (1931 through 1935 birth years) that began school prior to *Mendez*, they amount to 10.2 versus 12 years of schooling, respectively. The same difference is much smaller for younger cohorts (1941 through 1945 birth years) that began school after the *Mendez* ruling (11.7 versus 12.8 years of schooling, respectively). Thus, the difference-in-differences estimate using only these raw averages across groups suggests a substantial impact of desegregation on Hispanics of 0.7 years of completed schooling, or about 7 percent relative to the average in *HiSeg* counties among cohorts that began school completion outcomes, though only the difference-in-differences estimated impacts on years of schooling and junior high school completion outcomes are statistically significant at the 5% and 1% levels, respectively, in these summary statistics. The regression analysis will bolster this analysis by including additional controls for birth year and county fixed effects, as well as checking for robustness to pre-existing trends.

IV. EMPIRICAL STRATEGY

To analyze the effect of Mexican-American school desegregation on educational attainment, we estimate the following regression model by Ordinary Least Squares:¹⁹

(1)
$$Y_{ict} = \alpha + \pi \cdot (HiSeg_c \cdot PostMendez_t) + X_{ict} \cdot \beta + \mu_c + \delta_t + \varepsilon_{ict}$$

where the dependent variable, Y_{ict} , is an educational outcome for individual *i* from county *c* born in year *t* and the sample is limited to men and women from the 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 75% level for all 1940 counties) or very low (below the 25% level for all 1940 counties). *HiSeg_c* is an indicator variable for Hispanic to non-Hispanic population ratio population ratio in county *c* being above the 75% level of all California counties based on the 1940 complete count U.S. Census. *PostMendez_t* is a binary variable that is equal to 1 if individual *i* is born between the years

¹⁹ Note the empirical focus on Hispanics includes Mexican-Americans, as discussed in Section III.

1941 to 1945 (ages 2 to 6 at the time of the *Mendez* decision, i.e., fully exposed to desegregation for all school years), and zero otherwise. In the main results, the sample includes 1941-1945 birth years as well as individuals born 10 years prior, i.e., 1931-1935 birth years (ages 12-16 at the time of *Mendez*), and thus the primary comparison group consists of individuals who likely completed elementary school prior to *Mendez*. Note that omitting the intervening birth years (1936-1940), who would have been partially treated (ages 7 to 11, i.e., in elementary school, at the time of the *Mendez* decision), allows for a more precise comparison of the intervention, in the style of Duflo (2001). X_{ict} is a vector of control variables that includes an indicator for female, and indicators for 1990 and 2000 Census observations, respectively. Lastly, all regression specifications include county of residence fixed effects, μ_c , and birth year fixed effects, δ_t .²⁰ ε_{ict} is an i.i.d. error term, assumed to be independent within, but not necessarily across counties, so we cluster standard errors at the county level.

This difference-in-differences research design compares the educational attainment of younger Hispanic cohorts who were *fully treated* by the *Mendez* court ruling (cohorts born between 1941 and 1945) to individuals from older cohorts who started school prior to the *Mendez* ruling, based on their likelihood of attending segregated or non-segregated schools prior to the ruling (*HiSeg* versus *LoSeg* counties). One feature of the empirical approach is that it compares Hispanic cohorts that are relatively close in age (birth years 1941 through 1945 and birth years 1931 through 1935) and are thus likely to face similar circumstances aside from desegregation, allowing us to better isolate the impact of treatment. However, one concern with this strategy is that the comparison group might also be *partially treated* by the intervention, since the oldest among them would still be teenagers at the time of the *Mendez* ruling. Nevertheless, it is reasonable to expect that the impact on this comparison

²⁰ Note that the *Post Mendez* cohort dummy is collinear with the birth year fixed effects just as the *HiSeg* dummy is collinear with the county fixed effects and thus both are dropped from the model. Also note that if we dropped the fixed effects and instead included a post-Mendez dummy and high segregation county indicator in the regression model, that would give the same results as the last column of Table 1, which shows estimates that are very similar to our main results using county and birth year fixed effects. We prefer the specification with fixed effects because it allows us to control for variation across individual years and counties whereas the basic difference-in-differences design groups counties and birth cohorts into much more aggregated categories. Thus, the current specification allows us to address concerns that other differences across counties and/or across years are accounting for the changes we observe (as opposed to Mendez) and still estimate the differential impact of Mendez on educational attainment across high and low segregation counties.

group would be smaller relative to the treatment group if one expects greater returns from earlier educational investments²¹, and that the cost of switching schools would be greater for those who were already enrolled in school at the time of the *Mendez* decision. Thus, estimates using this comparison group likely represent a lower bound on the impact of the desegregation treatment. To obtain an upper bound on the impact of the treatment, subsequent analysis uses an older comparison group (birth years 1921 through 1930) that would likely have finished schooling at the time of the *Mendez* ruling, and thus would not have benefited from the desegregation treatment during their schooling years.

The identification strategy assumes that no other contemporaneous factors can explain the differences observed over this time period, after controlling for birth year fixed effects, which would account for shocks common to individuals across counties born in the same birth year (e.g., wars and national economic shocks), and county fixed effects, which would purge the estimates of time-invariant differences common to all individuals in a given county (e.g., poverty and ethnic attitudes). It also requires that individuals in the treatment (*HiSeg*) and comparison (*LoSeg*) groups would have maintained parallel trends in the absence of the treatment (desegregation around the time of *Mendez*). To bolster support for this assumption, as a robustness check, we run a placebo experiment where we replace the birth cohorts in the above analysis with cohorts who would all likely have completed schooling prior to *Mendez*, replacing the *PostMendez*_t indicator in equation (1) with a placebo cohort indicator (birth years 1926-1930) and where the comparison group is a slightly older cohort (birth years 1921-1925).

To assess the evidence for pre-existing trends, we also trace out differences in educational outcomes for individuals from *HiSeg* and *LoSeg* counties across a range of birth years that includes individuals who would likely have completed schooling prior to Mendez. At the same time, these event studies allow us to investigate the plausibility of partial impacts for individuals who were already in school at the time of Mendez. These point estimates are noisy due to the smaller sample sizes in isolating treatment effects by birth year in our sample, but provide some indication of stable trends prior to *Mendez* as well as

²¹ Indeed, there is a salient and robust literature in the social sciences on the short- and long-term impacts of investing in early childhood development (See, for example, Currie and Thomas 1995, 1999; Currie 2001; Garces, Thomas, and Currie 2002; Deming 2009; Elder and Lubotsky 2009; Bassok and Loeb 2015).

	Panel A: Hispanic Sample		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	0.880**	0.145***	0.093**
	(0.331)	(0.034)	(0.044)
Mean (y-variable)	11.17	0.88	0.60
Number of observations	5,621	5,621	5,621
R ²	0.10	0.09	0.07

Table 2: The Impact of Mendez v	. Westminster School Desegregation Decision on Educational Attainment
Free Free Free Free Free Free Free Free	

	Panel B: Placebo Sample (Hispanic Birth Cohorts 1921-1930)		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Placebo Post-Mendez Cohort	-0.319	-0.010	-0.133*
	(0.415)	(0.064)	(0.069)
Mean (y-variable)	9.17	0.70	0.36
Number of observations	3,423	3,423	3,423
R ²	0.07	0.08	0.04

	Panel C: Non-Hispanic White Sample		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	-0.386***	-0.009**	-0.025**
	(0.078)	(0.004)	(0.010)
Mean (y-variable)	13.44	0.99	0.90
Number of observations	27,812	27,812	27,812
R^2	0.05	0.01	0.02
Birth Cohort Fixed Effects	Yes	Yes	Yes
County Fixed Effects	Yes	Yes	Yes

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 75% level of all California counties based on 1940 full-count U.S. Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table 2. Sample is limited to men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 75% level for all 1940 counties: high segregation) or very low (below the 25% level for all 1940 counties: low segregation). Samples in panels A and C include only those individuals with birth cohorts between 1941 and 1945 (treatment group) or birth cohorts between 1931 and 1935 (comparison group). Sample in panel B includes only those individuals with birth cohorts between 1926 and 1930 (placebo treatment group) or birth cohorts between 1921 and 1925 (placebo comparison group). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10.

evidence for partial benefits accruing to Hispanics who were in school at the time of the *Mendez* ruling. While our primary research question focuses on the impact of desegregation on Hispanic educational outcomes, as an additional extension, we also investigate the impacts of desegregation on non-Hispanic whites using the research design laid out above, but where the sample is limited to non-Hispanic whites.

V. RESULTS AND DISCUSSION

Table 2 presents our results from estimating equation (1) on the Hispanic sample as well as alternative samples. As shown in column (1) of panel A, the difference-in-differences coefficient in the specification with years of education as the dependent variable is statistically significant at the 1% level and suggests that school desegregation increased years of education for Hispanics by 0.88 years. When compared with the average years of education for Hispanics in *HiSeg* counties prior to *Mendez* from Table 1 (10.23 years), this amounts to an increase of about 8.6 percent. This relatively large increase suggests that improvements in schooling may have led to increases in certification, so the remaining columns (2) and (3) of panel A analyze whether Hispanics were more likely to complete junior high (at least 8 years of schooling) and high school (12 years of schooling or more). These difference-in-differences coefficients are also statistically significant at the 1% and 5% levels, respectively, and suggest an increase in the probability of junior high school completion of 14.5 percentage points and high school completion of 9.3 percentage points for Hispanics, or 18.4 percent and 19.4 percent, respectively, relative to their baseline levels of educational attainment prior to *Mendez.*²²

The primary concern with our empirical strategy is that it relies on the assumption that Hispanics in *HiSeg* and *LoSeg* counties would have maintained parallel trends in the absence of the treatment. To bolster support for this assumption, panel B of Table 2 checks for pre-existing trends by testing whether Hispanics in *HiSeg* counties experienced any improvements in educational attainment relative to *LoSeg* counties in a sample of cohorts that completed schooling prior to *Mendez*—groups that should all be *too old* to benefit from

²² Results estimated separately for Hispanic men and women are shown in appendix Table A1. These results show positive impacts for both groups.



Figure 2: Event Study Analysis – Educational Attainment for Hispanics, Junior High School Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1940 counties: high segregation) or very low (i.e., below the 25% level for all 1940 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.

schooling desegregation. Thus, the *placebo treatment* cohort (birth years between 1926 and 1930) in the analysis shown in panel B of Table 2 is a cohort that is slightly younger than the *placebo comparison* cohort (birth years between 1921 and 1925), but where neither cohort should benefit from treatment. If our main results were purely driven by differential growth rates in educational attainment over time across both the *HiSeg* and *LoSeg* counties, then we should observe that the point estimates in the placebo analysis are also positive, otherwise this evidence casts doubt on the spurious interpretation of the main results. Indeed, what we observe is that there are no statistically significant differences in years of education (column 1 of panel B) and the junior high school outcome (column 2 of panel B), and only a marginally statistically significant difference in the high school outcome (column 3 of panel B), but where all coefficients are negative. This confirms that, after accounting for the controls in the regression model, there was no pre-existing positive growth trend in *HiSeg* counties relative to *LoSeg* counties prior to desegregation, and is thus consistent with the causal interpretation of the desegregation treatment shown in panel A.

Another way to validate the use of our identification strategy is to present a full event study analysis showing the difference-in-differences coefficient estimates for individual birth years, that is, replacing the *PostMendez*_t indicator in equation (1) with a birth year indicator. This exercise can also be used to gauge whether there is evidence for partial treatment for cohorts who may have begun schooling prior to *Mendez* but who may not have completed schooling by the time of desegregation. In practice, this exercise is challenging due to small sample sizes at the single birth year level, however, we present the event study graph for the junior high school outcome in Figure 2.²³ Note also that this graph covers a large range of birth years, covering 1921-1945, where the reference category is the 1931 birth year (age 16 at the time of the *Mendez* decision). Since these graphs go from segregated to desegregated eras and consider both fully treated cohorts (i.e., began school after the *Mendez* decision), partially treated cohorts (i.e., in school during the *Mendez* decision), and untreated cohorts, it is perhaps unsurprising that we observe what appears to be three emergent areas of the graph. For the oldest cohorts, who were in their 20s at the time of *Mendez*, we see a relative disadvantage for Hispanics in *HiSeg* counties relative to *LoSeg*

²³ The years of education and high school outcome event study analyses are qualitatively similar and can be found in the appendix (see Figures A1 and A2).

counties and this trend is relatively stable over time. For those cohorts who were in their teens (ages 12-16) at the time of *Mendez*, and who might be thought of as partially treated by desegregation, we see a stable trend around zero relative to the reference category. The latter group is the comparison group in the main results (shown in panel A of Table 2), and is marked as "Old Group" in Figure 2. Finally, for those birth years that were fully exposed to desegregation, as a consequence of starting school after *Mendez* (our *PostMendez*_t cohort in equation (1), noted as "Young Group" (ages 2-6) in Figure 2), we see a positive impact relative to the reference category. While confidence intervals are wide due to small sample sizes at the birth-year level, we view this event study graph as consistent with the overall research design in showing that the difference across *HiSeg* and *LoSeg* counties followed a stable pattern until the advent of desegregation. It also serves to highlight the likelihood of partial treatment exposure for cohorts in school at the time of the *Mendez* decision, and thus suggests that the estimates reported in Table 2 are likely to be a conservative lower bound estimate of the impact of *Mendez*.²⁴

As an extension, we examine whether there are any spillover effects of desegregation using the same identification strategy and regression model in equation (1), but now limiting the sample to non-Hispanic whites.²⁵ These results, shown in Table 2 of panel C, suggest statistically significant decreases associated with desegregation for non-Hispanic whites in terms of educational attainment, junior high school completion, and high school completion (coefficient estimates -0.39, -0.01, -0.03, respectively), but point estimates are much smaller in magnitude relative to the positive impacts for Hispanics shown in panel A. This relative decline in educational outcomes for non-Hispanic whites in high segregated counties

²⁴ Event study graphs for Hispanic and non-Hispanic whites when *HiSeg* and *LoSeg* counties are defined using the 1930 Census are shown in appendix Figures A3 through A8, and show positive impacts of *Mendez* on Hispanics and negative impacts on non-Hispanic whites in *HiSeg* counties as well as evidence of stable trends prior to *Mendez*. Note that this implies a different set of counties than the main results and San Diego county is included in the High Segregation counties in that alternative set.

²⁵ Note that any degree of ethnic attrition, whereby individuals were classified as Hispanic at school-age, but no longer identify as Hispanic in later years, would likely yield an upward bias in the estimates for non-Hispanic whites, assuming those individuals would have responded to the desegregation treatment by increasing their educational attainment. This is a reasonable assumption, given evidence that Mexican-American ethnic attritors (classified by measures of ancestry, but not by self-identification) have higher educational attainment than the population that identifies as Mexican-American (Duncan and Trejo 2011). Thus, to the extent that we find negative estimates for the non-Hispanic white population, they are conservative as to the likely impact of desegregation on this group.

Table 3: Impact of Mendez v. Westminster with an Alternative Comparison Group, Post-Mendez Cohorts (1941-1945 Birth Cohorts) Relative to 1921-1930 Birth Cohorts

	Panel A: Hispanic Sample			
	(1)	(2)	(3)	
	Years of education	Jr. High School	High School	
High Segregation x Post-Mendez Cohort	1.900***	0.229***	0.184***	
	(0.330)	(0.041)	(0.030)	
Mean (y-variable): 1921-1930 Birth Cohorts	9.17	0.70	0.36	
Number of observations	6,803	6,803	6,803	
R^2	0.17	0.14	0.13	
		Panel B: Hispanic Men		
	(1)	(2)	(3)	
	Years of education	Jr. High School	High School	
High Segregation x Post-Mendez Cohort	2.011***	0.212***	0.125**	
	(0.339)	(0.046)	(0.045)	
Mean (y-variable): 1921-1930 Birth Cohorts	9.44	0.72	0.39	
Number of observations	3,218	3,218	3,218	
R ²	0.17	0.13	0.14	
	Ι	Panel C: Hispanic Women		
	(1)	(2)	(3)	
	Years of education	Jr. High School	High School	
High Segregation x Post-Mendez Cohort	1.937***	0.244***	0.244***	
	(0.420)	(0.054)	(0.043)	
Mean (y-variable): 1921-1930 Birth Cohorts	8.95	0.67	0.33	
Number of observations	3,585	3,585	3,585	
R ²	0.16	0.15	0.13	
Birth Cohort Fixed Effects	Yes	Yes	Yes	
County Fixed Effects	Yes	Yes	Yes	

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 75% level of all California counties based on 1940 full-count Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female (panel A only), and in all panels an indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table 3. Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses in California whose birth cohorts are between 1941 and 1945 (treatment group) and birth cohorts between 1921 and 1930 (comparison group), and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 75% level for all 1940 counties: high segregation) or very low (below the 25% level for all 1940 counties: low segregation). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10.

compared with non-Hispanic whites in low segregated counties after school desegregation mirrors patterns seen in Figure 1 as discussed above. Moreover, these findings would be consistent with school resources being shifted from non-Hispanic whites to Hispanics after school desegregation in high segregated counties, but absent data on school resources within schools explicitly identified as Mexican or white-only, we cannot definitively say whether this is the mechanism driving the results from above. Nevertheless, the fact that estimated impacts for non-Hispanic whites are much smaller in magnitude than for Hispanics suggest that desegregating schools improved schooling for Hispanics with relatively small impacts on non-Hispanic whites.

Finally, to provide an upper bound on our estimates of the impact of school desegregation on Hispanics, we examine treatment impacts of school desegregation relative to the older cohorts who completed schooling prior to *Mendez* by estimating the regression model in equation (1) with an alternative comparison group (birth years 1921-1930). These results provide an upper bound on the impact of desegregation on the educational attainment of Hispanics which can be compared with the lower bound of treatment impacts found by using the *partially treated* comparison group (birth years 1931-1935) presented in Table 2. The results of this alternative analysis are shown in Table 3 and show that, as expected, treatment impacts are much larger when we consider a comparison group that should not have been affected by Mendez at all. This amounts to 1.9 years of additional years of education, a 23 percentage point increase in the likelihood of completing junior high, and an 18 percentage point increase in the likelihood of competing high school. Since these can be compared to lower baseline averages for educational attainment in *HiSeg* counties for older cohorts (noted in Table 3), they represent even larger effective increases in educational attainment for treated groups. Panels B and C of Table 3 present these results separately by gender and show statistically significant impacts of desegregation on both men and women. Thus, these estimates suggest similar impacts for men and women with regards to educational attainment, after taking into account that women in older cohorts have lower educational attainment than men, but suggest somewhat larger impacts on junior high and high school completion rates for women.²⁶

²⁶ Extended results for non-Hispanic whites show negative impacts for men and women (see appendix Table A2), similar to the results in panel C of Table 2. Event study analyses using alternative definitions of *HiSeg* and

VI. CONCLUSION

This paper provides the first quantitative analysis of the effect of Mexican-American school desegregation by examining the impact of the 1947 *Mendez v. Westminster* decision on long-run educational outcomes of Hispanics in California. Despite a lack of official data on segregation, by comparing individuals in counties which were more likely to have segregated schools with individuals in counties which were less likely to have segregated schools across birth cohorts that started school after *Mendez* relative to birth cohorts that started school after *Mendez* relative evidence on its impacts. In particular, the results here show that desegregation significantly improved the educational attainment of Mexican-Americans, with some evidence of partial benefits for students already in school after *Mendez*. At the same time, similar analysis on non-Hispanic white groups suggests there may have been some negative impacts associated with school desegregation, which would be consistent with distributing resources more equitably across racial and ethnic groups after desegregation.

By providing the first quantitative estimates of desegregating Mexican-American schools, this work has made a significant contribution to the literature, but is limited in a number of ways. First, we are not able to directly tie individuals to schools, and most notably, schools and classrooms lack official designations as "Mexican/Mexican-American only" or "white only" even when this was the practice. Ideally, individual data would indicate the school and classroom of attendance over time, as well as the extent of segregation in schools and classrooms, along with educational quality measures such as teacher education, experience, wages, and student-teacher ratios. Unfortunately, lacking systematic historical data on these classifications and associated measures of resource allocation makes such investigations difficult, and require a more nuanced approach. Nevertheless, pinning down the mechanisms underlying the link between desegregation and greater equity in education is an important area for future research, especially given the increased levels of racial and ethnic segregation experienced by students today.

LoSeg counties based on the 1930 Census population distribution also suggest negative impacts for non-Hispanic whites (see appendix Figures A6 through A8).

REFERENCES

- Antman, Francisca, Brian Duncan, and Stephen J. Trejo. 2016. "Ethnic Attrition and the Observed Health of Later Generation Mexican Americans" *The American Economic Review*, 106(5): 467-71.
- Antman, Francisca, Brian Duncan, and Stephen J. Trejo. 2020. "Ethnic Attrition, Assimilation, and the Measured Health Outcomes of Mexican Americans" *Journal of Population Economics*, 33(4): 1499-1522.
- Arriola, Christopher. 1995. "Knocking on the Schoolhouse Door: Mendez v. Westminster, Equal Protection, Public Education, and Mexican Americans in the 1940's." *La Raza Law Journal*, 8(2): 166-207.
- Ashenfelter, Orley, William J. Collins, and Albert Yoon. 2006. "Evaluating the role of Brown v. Board of Education in school equalization, desegregation, and the income of African Americans." *American Law and Economics Review* 8(2): 213-248.
- Bassok, D, and Loeb, S. 2015. Early childhood and the achievement gap. Handbook of research in Education Finance and Policy, 510-528.
- Bowman, Kristi L. 2001. "The New Face of Desegregation." Duke Law Journal, 50: 1751-1808.
- California State Department of Education. 1932. *Biennial Report of the State Department of Education For the School Years Ending June 30, 1929, and June 30, 1930, Part Two.* State of California Division of Research and Statistics.
- Card, David, and Alan B. Krueger. 1992. "School Quality and Black-White Relative Earnings: A Direct Assessment." *The Quarterly Journal of Economics* 107(1): 151-200.
- Carruthers, Celeste K. and Marianne H. Wanamaker. 2017. "Returns to School Resources in the Jim Crow South." *Explorations in Economic History* 64:104-110.
- Carter, Thomas P. 1970. *Mexican Americans in School: A History of Educational Neglect*. New York: College Entrance Examination Board.
- Cascio, Elizabeth, Nora Gordon, Ethan Lewis, and Sarah Reber. 2010. "Paying for Progress: Conditional Grants and the Desegregation of Southern Schools." *Quarterly Journal of Economics*, 125(1): 445-82.
- Cook, Lisa, Trevon Logan, and John Parman. 2017. "Racial Segregation and Southern Lynching." National Bureau of Economic Research Working Paper No. 23813.
- Currie, Janet. 2001. "Early Childhood Education Programs." *Journal of Economic Perspectives*, 15 (2): 213-238.

- Currie, Janet and Duncan Thomas. 1995. "Does Head Start Make a Difference?" *The American Economic Review*. 85:3, pp. 341–64.
- Currie, Janet and Duncan Thomas. 1999. "Does Head Start Help Hispanic Children?" *Journal* of Public Economics. 74:2, pp. 235–62
- Deming, David, "Early childhood intervention and life-cycle skill development: Evidence from Head Start," *American Economic Journal: Applied Economics*, 2009, 1(3), 111–134.
- Donato, Rubén, Martha Menchaca, and Richard R. Valencia. 1991. "Segregation, Desegregation, and Integration of Chicano Students: Problems and Prospects." In *Chicano School Failure and Success: Research and Policy Agendas for the 1990s.* Ed. Valencia, Richard R. New York: The Falmer Press.
- Duflo, Esther. 2001. "Schooling and Labor Market Consequences of School Construction in Indonesia: Evidence from an Unusual Policy Experiment." *The American Economic Review*. 91(4): 795-813.
- Duncan, Brian and Stephen J. Trejo. 2011. "Intermarriage and the Intergenerational Transmission of Ethnic Identity and Human Capital for Mexican Americans." *Journal of Labor Economics*, 29(2): 195-227.
- Elder, Todd E., and Darren H. Lubotsky. 2009. Kindergarten entrance age and children's achievement: Impacts of state policies, family background, and peers. *The Journal of Human Resources* 44(3):641–683.
- Garces, Eliana, Duncan Thomas, and Janet Currie. 2002. "Longer-term effects of Head Start," *The American Economic Review*, 92(4), 999–1012.
- Gonzalez, Gilbert G. 1985. "Segregation of Mexican children in a southern California city: The legacy of expansionism and the American southwest." *The Western Historical Quarterly* 16(1): 55-76.
- Gonzalez, Gilbert G. 1990. *Chicano Education in the Era of Segregation*. Philadelphia: The Balch Institute Press.
- Guryan, Jonathan. 2004. "Desegregation and Black Dropout Rates." *The American Economic Review*, 94(4): 919-43.
- Hendrick, Irving G. 1975. "Public Policy toward the Education of Non-White Minority Group Children in California, 1849-1970." Final Report, National Institute of Education Project No. NE-G-00-3-0082. University of California Riverside, School of Education.
- Hendrick, Irving G. 1977. "The Education of Non-Whites in California, 1849-1970." R and E Research Associates, San Francisco, CA.

- Johnson, Rucker C. 2011. "Long-Run Impacts of School Desegregation and School Quality on Adult Attainments." National Bureau of Economic Research Working Paper No. 16664.
- Kandil, Caitlin Yoshiko. 2016. "Mendez vs. segregation: 70 years later, famed case 'isn't just about Mexicans. It's about everybody coming together'." *Los Angeles Times*. April 17, 2016.
- Logan, Trevon D. and John M. Parman. 2017. "The National Rise in Residential Segregation." *Journal of Economic History.* 77(1): 127-70.
- Margo, Robert A. 1990. "The Impact of Separate-but-Equal" in *Race and Schooling in the South, 1880-1950: An Economic History*. Chicago: University of Chicago Press, 68-86.
- Meraji, Shereen Marisol. 2014. "Before `Brown v. Board,' Mendez Fought California's Segregated Schools." *National Public Radio.* May 16, 2014.
- Motel, Seth and Eileen Patten. 2012. "The 10 Largest Hispanic Origin Groups: Characteristics, Rankings, Top Counties." Pew Research Center. Available at http://www.pewhispanic.org/2012/06/27/the-10-largest-hispanic-origin-groupscharacteristics-rankings-top-counties/
- Orfield, Gary and Jongyeon Ee. 2014. "Segregating California's Future: Inequality and Its Alternative 60 Years after *Brown v. Board of Education*." Report of the Civil Rights Project, University of California Los Angeles.
- Reber, Sarah J. 2005. "Court-Ordered Desegregation: Successes and Failures Integrating American Schools since Brown v Board of Education." *The Journal of Human Resources*, 40(3): 559-90.
- Reber, Sarah J. 2010. "School Desegregation and Educational Attainment for Blacks." *The Journal of Human Resources*, 45(4): 893-914.
- Ruggles, Steven, Katie Genadek, Ronald Goeken, Josiah Grover, and Matthew Sobek. *Integrated Public Use Microdata Series: Version 6.0* [dataset]. Minneapolis: University of Minnesota, 2015. http://doi.org/10.18128/D010.V6.0.

Treaty of Guadalupe Hidalgo. 1848. U.S.-Mex. Feb. 2, 1848.

Valencia, Richard R. 2005. "The Mexican American Struggle for Equal Educational Opportunity in Mendez v. Westminster: Helping to Pave the Way for Brown v. Board of Education." *Teacher's College Record*. 107(3): 389-423.

Westminster School Dist. of Orange County v. Mendez, 161 F.2d 774 (9th Cir. 1947).

Wollenberg, Charles. 1976. "All Deliberate Speed: Segregation and Exclusion in California Schools, 1855-1975." University of California Press.

THE LONG-RUN IMPACTS OF MEXICAN-AMERICAN SCHOOL DESEGREGATION

ONLINE APPENDIX December 27, 2021

Francisca M. Antman University of Colorado Boulder francisca.antman@colorado.edu

> Kalena E. Cortes Texas A&M University kcortes@tamu.edu

	Panel A: Hispanic Sample (same as Table 2, Panel A)		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	0.880**	0.145***	0.093**
	(0.331)	(0.034)	(0.044)
Mean (y-variable)	11.17	0.88	0.60
Number of observations	5,621	5,621	5,621
R ²	0.10	0.09	0.07
		Panel B: Hispanic Men	
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	1.084***	0.156***	0.134
	(0.297)	(0.029)	(0.079)
Mean (y-variable)	11.55	0.90	0.65
Number of observations	2,718	2,718	2,718
R ²	0.10	0.09	0.06
	Pa	anel C: Hispanic Women	
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	0.719*	0.135**	0.065
	(0.372)	(0.059)	(0.044)
Mean (y-variable)	10.81	0.87	0.56
Number of observations	2,903	2,903	2,903
R ²	0.09	0.09	0.07
Birth Cohort Fixed Effects	Yes	Yes	Yes
County Fixed Effects	Yes	Yes	Yes

Table A1: Extended Results By Gender - Impact of Mendez v. Westminster on Hispanic Educational Attainment for Post-Mendez Cohorts (1941-1945 Birth Cohorts) Relative to 1931-1935 Birth Cohorts

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 75% level of all California counties based on 1940 full-count Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female (panel A only), and in all panels an indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table A1. Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses in California whose birth cohorts are between 1941 and 1945 (treatment group) and birth cohorts between 1931 and 1935 (comparison group), and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 75% level for all 1940 counties: high segregation) or very low (below the 25% level for all 1940 counties: low segregation). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10.

	Panel A: Non-Hispanic White Sample		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	-0.304**	-0.004	-0.030**
	(0.116)	(0.006)	(0.012)
Mean (y-variable)	13.25	0.98	0.87
Number of observations	36,029	36,029	36,029
R ²	0.06	0.01	0.03
	Panel B: Non-Hispanic White Men		
	Pane	l B: Non-Hispanic White M	len
	Pane	l B: Non-Hispanic White M (2)	len (3)
	Pane (1) Years of education	l B: Non-Hispanic White M (2) Jr. High School	len (3) High School
High Segregation x Post-Mendez Cohort	Pane (1) Years of education -0.314*	l B: Non-Hispanic White M (2) Jr. High School -0.002	Ien (3) High School -0.032
High Segregation x Post-Mendez Cohort	Pane (1) Years of education -0.314* (0.166)	l B: Non-Hispanic White M (2) Jr. High School -0.002 (0.007)	Ien (3) High School -0.032 (0.020)
High Segregation x Post-Mendez Cohort Mean (y-variable)	Pane (1) Years of education -0.314* (0.166) 13.54	l B: Non-Hispanic White M (2) Jr. High School -0.002 (0.007) 0.98	Ien (3) High School -0.032 (0.020) 0.87
High Segregation x Post-Mendez Cohort Mean (y-variable) Number of observations	Pane (1) Years of education -0.314* (0.166) 13.54 17,398	l B: Non-Hispanic White M (2) Jr. High School -0.002 (0.007) 0.98 17,398	Ien (3) High School -0.032 (0.020) 0.87 17,398

Table A2: Impact of Mendez v. Westminster on Non-Hispanic Whites with Alternative Comparison Group, Post-Mendez Cohorts (1941-1945 Birth Cohorts) Relative to 1921-1930 Birth Cohorts

	Panel	Panel C: Non-Hispanic White Women		
	(1)	(2)	(3)	
	Years of education	Jr. High School	High School	
High Segregation x Post-Mendez Cohort	-0.311***	-0.006	-0.028**	
	(0.093)	(0.008)	(0.010)	
Mean (y-variable)	12.99	0.99	0.88	
Number of observations	18,631	18,631	18,631	
R ²	0.04	0.01	0.02	
Birth Cohort Fixed Effects	Yes	Yes	Yes	
County Fixed Effects	Yes	Yes	Yes	

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 75% level of all California counties based on 1940 full-count Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female (panel A only), and in all panels an indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table A2. Sample is limited to non-Hispanic white men and women from 5% samples of 1980, 1990, and 2000 Censuses in California whose birth cohorts are between 1941 and 1945 (treatment group) and birth cohorts between 1921 and 1930 (comparison group), and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 75% level for all 1940 counties: high segregation) or very low (below the 25% level for all 1940 counties: low segregation). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10. 30

	Panel A: Hispanic Sample		
	(1) Years of education	(2) Ir High School	(3) High School
High Segregation x Post-Mendez Cohort	0.702**	0.114***	0.089**
	(0.302)	(0.032)	(0.041)
Mean (y-variable)	11.24	0.89	0.61
Number of observations	12,742	12,742	12,742
R ²	0.07	0.06	0.05
	Panel B: Pla	cebo Sample (Hispanic Birth C	ohorts 1921-1930)
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Placebo Post-Mendez Cohort	-0.398	-0.015	-0.143*
	(0.399)	(0.056)	(0.069)
Mean (y-variable)	9.51	0.74	0.37
Number of observations	8,674	8,674	8,674
R ²	0.04	0.05	0.02
		Panel C: Non-Hispanic White	Sample
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	-0.420***	-0.012***	-0.036***
	(0.077)	(0.003)	(0.011)
Mean (y-variable)	13.59	0.99	0.90
Number of observations	51,191	51,191	51,191
R^2	0.05	0.01	0.02
Birth Cohort Fixed Effects	Yes	Yes	Yes
County Fixed Effects	Yes	Yes	Yes

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 67% level of all California counties based on 1940 full-count U.S. Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table A3. Sample is limited to men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 67% level for all 1940 counties: high segregation) or very low (below the 33% level for all 1940 counties: low segregation). Samples in panels A and C include only those individuals with birth cohorts between 1941 and 1945 (treatment group) or birth cohorts between 1931 and 1935 (comparison group). Sample in panel B includes only those individuals with birth cohorts between 1926 and 1930 (placebo treatment group) or birth cohorts between 1921 and 1925 (placebo comparison group). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10. 31

	Panel A: Hispanic Sample		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	1.621***	0.185***	0.175***
	(0.296)	(0.039)	(0.027)
Mean (y-variable)	10.57	0.83	0.52
Number of observations	16,162	16,162	16,162
R ²	0.13	0.10	0.11
	Panel B: Hispanic Men		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	1.687***	0.170***	0.115***
	(0.294)	(0.043)	(0.040)
Mean (y-variable)	10.88	0.85	0.55
Number of observations	7,596	7,596	7,596
R ²	0.13	0.09	0.12
	Panel C: Hispanic Women		
	(1)	(2)	(3)
	Years of education	Jr. High School	High School
High Segregation x Post-Mendez Cohort	1.706***	0.197***	0.239***
	(0.374)	(0.050)	(0.041)
Mean (y-variable)	10.30	0.82	0.49
Number of observations	8,566	8,566	8,566
R^2	0.12	0.11	0.10
Birth Cohort Fixed Effects	Yes	Yes	Yes
County Fixed Effects	Yes	Yes	Yes

Table A4: Impact of Mendez v. Westminster with Alternative High and Low Segregation Definitions, Post-Mendez Cohorts (1941-1945 Birth Cohorts) Relative to 1921-1930 Birth Cohort Comparison Group

Notes: High segregation (HiSeg) indicates that the Hispanic to non-Hispanic population ratio is above the 67% level of all California counties based on 1940 full-count U.S. Census. Post-Mendez Cohort is an indicator for birth year being 1941 or later. Other controls include indicator for female (panel A only), indicator for 1990 Census observation and indicator for 2000 Census observation, respectively, in addition to fixed effects noted in Table A4. Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (above the 67% level for all 1940 counties: high segregation) or very low (below the 33% level for all 1940 counties: low segregation). Samples in panels A and C include only those individuals with birth cohorts between 1941 and 1945 (treatment group) or birth cohorts between 1931 and 1935 (comparison group). Sample in panel B includes only those individuals with birth cohorts between 1921 and 1925 (placebo comparison group). Robust standard errors, clustered at county level, in parentheses. Statistical significance levels: *** p<0.01, ** p<0.05, * p<0.10.



Figure A1: Event Study Analysis – Educational Attainment for Hispanics, Years of Education Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1940 counties: high segregation) or very low (i.e., below the 25% level for all 1940 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A2: Event Study Analysis – Educational Attainment for Hispanics, High School Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1940 counties: high segregation) or very low (i.e., below the 25% level for all 1940 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A3: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Hispanics, Years of Education Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A4: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Hispanics, Junior High School Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A5: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Hispanics, High School Outcome

Notes: Sample is limited to Hispanic men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A6: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Non-Hispanic Whites, Years of Education Outcome

Notes: Sample is limited to non-Hispanic white men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A7: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Non-Hispanic Whites, Junior High School Outcome

Notes: Sample is limited to non-Hispanic white men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.



Figure A8: Event Study Analysis with 1930 County Segregation Data – Educational Attainment for Non-Hispanic Whites, High School Outcome

Notes: Sample is limited to non-Hispanic white men and women from 5% samples of 1980, 1990, and 2000 Censuses who were born in California and who reside in a county where the Hispanic to non-Hispanic population ratio was either very high (i.e., above the 75% level for all 1930 counties: high segregation) or very low (i.e., below the 25% level for all 1930 counties: low segregation). Graph shows the difference-in-differences coefficient estimate on birth year interacted with high segregation county indicator. All regression models also include birth year fixed effects, county fixed effects, indicator for female, indicator for 1990 Census observation and indicator for 2000 Census observation, respectively. Shaded areas indicate the 90% confidence intervals, where standard errors are clustered at county level.